Author's response to reviews

Title: Efficacy and safety of tigecycline versus levofloxacin for community-acquired pneumonia

Authors:

Cristina Tanaseanu (cristina.tanaseanu@gmail.com)
Slobodan Milutinovic (slobodan.milutinovic@bol-svduh.htnet.hr)
Petre I Calistru (pcalistru@yahoo.com)
Janos Strausz (strausz@koranyi.hu)
Marius Zoulbas (zolubas@omni.lt)
Valeriy Chernyak (schevchenko@neocm.com)
Nathalie Dartois (dartoin@wyeth.com)
Nathalie Castaing (castain@wyeth.com)
Hassan Gandjini (gandjih@wyeth.com)
C A Cooper (cooperc2@wyeth.com)

Version: 2 Date: 29 July 2009

Author's response to reviews: see over
Dear Editors,

Thank you for considering our manuscript entitled “Efficacy and safety of tigecycline versus levofloxacin for community-acquired pneumonia”, for publication in *BMC Pulmonary Medicine*.

Per your suggestion, the “Competing Interests” section of the manuscript has been revised to specify that Wyeth Research funded this study, and the overall formatting has been updated per the provided checklist. No additional revisions were suggested or requested by the referees.

Thank you.